University of Waterloo STAT 332 – W. H. Cherry

#2.91a

Figure 2.12. UNSTRATIFIED POPULATIONS: Estimating the Difference of Two Proportions Section headings to provide a content overview

One-Stage EPSWOR of Individual Elements

1. Case 1: Estimating^−P_1 −^−P_2 when^−P_1 and^−P_2 Involve Different Populations 2. Case 2: Estimating^−P_1 −^−P_2 when^−P_1 and^−P_2 Involve the Same Population

Example 2.12.1 – Solution 1 Example 2.12.1 – Solution 2 Example 2.12.2 – Solution 1 Example 2.12.2 – Solution 2 3. Appendix 1: Newspaper Article for Examples 2.12.1 and 2.12.2 4. Appendix 2: Newspaper Article for Example 2.12.1 Fourteen equations

2023-08-20